Author's response to reviews

Title: Counting on commitment; the quality of primary care-led diabetes management in a system with minimal incentives

Authors:

Sheena M Mc Hugh (s.mchugh@ucc.ie)
Paul Marsden (paul.marsden@hse.ie)
Carmel Brennan (carmel.brennan@hse.ie)
Katie Murphy (Katie.murphy@ucc.ie)
Celine Croarkin (celine.croarkin@hse.ie)
Joe Moran (j.moran@ucc.ie)
Velma Harkins (vharkins@eircom.net)
Ivan J Perry (i.perry@ucc.ie)

Version: 5 Date: 6 December 2011

Author's response to reviews: see over
Re: MS 2111439574536097 – Counting on commitment; the quality of primary care-led diabetes management in a system with minimal incentives.

Dear Editor,

Thank you for inviting us to finalise our paper. The remaining minor revision regarding ethical approval has now been addressed in the methods section of the revised manuscript (attached). As requested, the following sentence has been inserted:

“The study was based on secondary data analysis of anonymised data which had previously been published and disseminated in individual audit reports from each of the three initiatives therefore ethical approval was not sought for the collated secondary analysis. The data was originally collected for the purpose of service evaluation which did not require patient consent.”

We look forward to hearing from you.

Best wishes

Sheena Mc Hugh (corresponding author)

HRB PhD Scholar in Health Services Research.

Department of Epidemiology and Public Health,
Brookfield Health Sciences Complex,
University College Cork,
College Rd.,
Cork,
Ireland.